Author’s response to reviews

Title: Rare cause of severe anemia due to pyogenic granuloma in the jejunum

Authors:
Shun-ichi Misawa (96084sm@jichi.ac.jp)
Hiroto Sakamoto (m05041hs@jichi.ac.jp)
Akira Kurogochi (159735@hp-hata.com)
Yasushi Kirii (yasukiri@jichi.ac.jp)
Shinichiro Nakamura (139882@hp-hata.com)
Tomoko Misawa (96014to@jichi.ac.jp)
Suguru Yoneda (119865@hp-hata.com)
Mari Hirano (159905@hp-hata.com)
Osamu Owa (0809@hp-hata.com)
Hiroyuki Takagi (0344@hp-hata.com)
Hiroyoshi Ota (hohta@shinshu-u.ac.jp)

Version: 1 Date: 05 Sep 2015

Author’s response to reviews

Thank you for reviewing our manuscript. I changed manuscript and abstract as follows.

Reviewer #1:

1. There is a non-sentence in the Abstract; all sentences should contain a verb.

2. Reference 4 is incomplete

   1. I changed abstract sentences and corrected grammatical error.

   2. Reference 4 was completed.
Reviewer #2: Please check grammatical error in abstract.

1. I changed abstract sentences and corrected grammatical error.

I hope our case report will contribute to your journal.